Author’s response to reviews

Title: First line in psychiatric emergency: Pre-hospital emergency protocol for mental disorders in Iran

Authors:

Fatemeh Shirzad (Shirzad.f@iums.ac.ir)
Fateme Hadi (Hadi.f@iums.ac.ir)
Seyede Salehe Mortazavi (salehe.mortazavi@gmail.com)
Maryam Biglari (dr.maryam54@gmail.com)
Hassan Noori Sari (hassannourisari@yahoo.com)
Zeinab Mohammadi (m.mohammady35@gmail.com)
Mehrdad Kazemzade Atoofi (atoofi.m@iums.ac.ir)
Seyed Vahid Shariat (vahidshariat@gmail.com)

Version: 2 Date: 24 Jan 2020

Author’s response to reviews:

Dear editorial manager of BMC Emergency Medicine,

This is the responds to the reviewer comments

1- The numbers of screened, included and excluded studies was added. (page 5 line 3)
2- table 1: The sample size (participants or papers) of each study was added
3- quality assessment of the included paper was explained. (page 5 line 4)
4- English proofreading, especially for the newly added texts was done.
5- Fonts throughout the manuscript, especially the newly added text were checked.

Sincerely,